Author's response to reviews

Title: Overexpression of YAP 1 contributes to progressive features and poor prognosis of human urothelial carcinoma of the bladder

Authors:

Jian-Ye Liu (liujiang@sysucc.org.cn)
Yong-Hong Li (liyongh@sysucc.org.cn)
Huan-Xin Lin (linhx@sysucc.org.cn)
Yi-Ji Liao (liaoy@sysucc.org.cn)
Shi-Juan Mai (maishj@sysucc.org.cn)
Zhou-Wei Liu (liuzhw@sysucc.org.cn)
Zhi-Ling Zhang (zhangzhl@sysucc.org.cn)
Li-Juan Jiang (jianglij@sysucc.org.cn)
Jia-Xing Zhang (zhangjx@sysucc.org.cn)
Hsiang-Fu Kung (kunghfsysucc@163.com)
Yi-Xin Zeng (zengyx@sysucc.org.cn)
Fang-Jian Zhou (zhoufj@sysucc.org.cn)
Dan Xie (xiedan@sysucc.org.cn)

Version: 2 Date: 26 March 2013

Author's response to reviews: see over
1. Page 6, lines 7-8: “Institute Research Medical Ethics Committee of Sun Yat-Sen University Cancer Center granted approval for this study.” changed to “Prior patients' consent and approval from the Institutional Research Medical Ethics Committee of Sun Yat-Sen University Cancer Center were obtained for the use of these clinical materials for research purposes.”